Author's response to reviews

Title: Lessons learned from implementation of a demonstration program to reduce the burden of anaemia and hookworm in women in Yen Bai Province, Viet Nam

Authors:

Tran Q Phuc (tguangphuc@yahoo.com)
Seema Mhrshahi (seemam@unimelb.edu.au)
Gerard J Casey (gcasey@unimelb.edu.au)
Luong B Phu (phuytyb@yahoo.com.vn)
Nong T Tien (nongthitien@yahoo.com)
Sonia R Caruana (sonia2rose@yahoo.com.au)
Tran D Thach (indthach@yahoo.com)
Antonio Montresor (montresora@vtn.wpro.who.int)
Beverley-Ann Biggs (babiggs@unimelb.edu.au)

Version: 8 Date: 16 July 2009

Author's response to reviews:

Final proofs with minor punctuation corrections.